



Article Outsourcing Strategies of Established Firms and Sustainable Competitiveness: Medical Device Firms

Byungjoo Paek, Joohyun Kim, Joonyoung Park and Heesang Lee *

Graduate School of Management of Technology, Sungkyunkwan University, Chunchundong 300, Suwon 440-746, Korea

* Correspondence: leehee@skku.edu; Tel.: +82-31-290-7604

Received: 28 June 2019; Accepted: 13 August 2019; Published: 22 August 2019



Abstract: Successful outsourcing strategy is a key enabler of sustainable competitiveness of established firms engaged in technology-intensive business. This study proposes a research framework of a successful outsourcing process composed of resources, dynamic capabilities and created values, from a review of relevant literature and in-depth case study of four leading medical device firms. The research results specify four outsourcing types according to the characteristics of outsourced resources, and their alignment with resources owned by the outsourcer. Each type of outsourcing is positively affected by the outsourcer's dynamic capabilities, namely sensing, integrating capability, absorptive capability and technology transfer capability. In addition, the significance of the outsourcee's size according to each outsourcing type is also found. As a result of successful outsourcing experience, the outsourcer can reap the benefits of enhanced dynamic capabilities, i.e., market-oriented innovation, strategic flexibility, agility and increased efficiency, thereby creating a virtuous cycle of sustainable competitiveness in a volatile environment.

Keywords: outsourcing strategy; sustainable competitiveness; established firms; dynamic capabilities; medical device industry

1. Introduction

As modern firms are confronted with higher levels of complexity and uncertainty, it is significant for the firms to adjust their capabilities to cope with environmental change for sustainable competitive advantages [1]. In such unpredictable business environments, in order to enhance their competitiveness, many established firms continuously utilize external resources, as well as internal capabilities [2]. Outsourcing has been a typical strategy that takes advantage of the more advanced or lower-priced resources with greater efficiency, and thereby brings strategic advantages to the outsourcing firms [3–5]. Outsourcing strategies lead the firms to have a broad network of the best competencies linked by outsourcing contracts, from which necessary functions such as raw materials, components and systems can be outsourced. Through the networked organizations, the outsourcing firms can transform themselves into more flexible and agile organizational structures [6,7]. For these reasons, an increasing number of firms have adopted outsourcing strategies, seeking enhanced efficiency and competitiveness, and a topic of strategic outsourcing is increasingly widespread among modern organizations, thereby attracting great interest from organizational scholars, as well as firm management [3,6,8–10]. According to Transaction Cost Economics (TCE), the costs associated with economic exchange play a leading role in the outsourcing decision, and the transactions are made through the governance form requesting minimum costs [11,12]. From the perspective of TCE, corporate management decides outsourcing to minimize transaction costs, in the case that outsourcing can be more cost effective than in-house development [13,14]. As it is increasingly difficult for firms to keep the boundaries of their core competences stable for long, the purposes of outsourcing have evolved from cost efficiency, e.g.,

in offshore manufacturing, to more strategic issues, such as flexibility, innovation, and sustainability, by reducing risk in a volatile environment [7,12,15]. Some studies support that a resource and competence-based perspective should complement TCE to analyze outsourcing activities, because a firm's inherent capability and strategic orientation, as well as cost efficiency, affect outsourcing decisions; and thus TCE alone is insufficient to explain the dynamics of governance in the outsourcing process [12,16,17].

Resource-Based View (RBV), which is another popular approach to outsourcing phenomena, focuses on the idiosyncratic resource owned by each firm as an effective factor to motivate outsourcing. The resource-based perspective of outsourcing posits that a firm can extend its boundary by building bridges to outsourcing partners, enabling access to immediately available technologies, while TCE considers a firm's boundary to be where they relinquish control over resources [13,18]. Outsourcing is considered as a strategic opportunity that can complement resources that are difficult to develop internally, and thereby generates synergies beyond combined value [3,17,19,20]. As a dynamic issue of outsourcing governance that changes over time beyond a static and restricted realm gains more attention along with changes in market needs, successful outsourcing practice between the outsourcer and outsourcing partner, i.e., the outsourcee, demands the outsourcer's dynamic capabilities, such as the sensing of valuable external resources, absorption of foreign resources, and integration of acquired resources with internal capacities [12,21,22]. The Dynamic Capability View (DCV) that is evolved from the traditional RBV, complementing its static nature, emphasizes the dynamic process of linking external resources or stimuli with internal capacities, and therefore is an important explanation of how corporate resources or capabilities affect outsourcing decisions, and how outsourcing practices of technology-intensive firms lead to long-term success [5,17,21,23]. According to DCV, a successful outsourcing experience creates positive value for the organization with enhanced dynamic capabilities, e.g., increased flexibility, agility, efficiency and innovativeness, which in turn leads to sustainable competitiveness of the firm [5,12] Nevertheless, few studies have focused on the established firm's dynamic capabilities in connection with successful outsourcing practice and sustainable competitiveness, and in addition, how a firm's inherent resource or capability affect outsourcing strategy according to the characteristics of resources and the firm size.

This study first suggests a research framework of successful outsourcing practice of established firms, comprised of resources, dynamic capabilities, and created values, which is based on a review of existing literature relevant to the topic of DCV and outsourcing. In sequence, a case study of the leading, established firms in the medical device business, which is technology-intensive and highly dependent on outsourcing, concretizes the proposed framework, by specifying outsourcing types characterized by a firm's retained resources and capabilities. In addition, the interrelationship between the outsourcing type and the outsourcee's size has been investigated. Although outsourcing is a strategically important field in the medical device industry that requires convergence with various ICT technologies, there is still a lack of relevant academic research. For an in-depth case study, content analysis for the relevant media articles from 2010 to 2018 has been conducted, in order to analyze detailed outsourcing events of four leading established firms in the industry: Medtronic (headquartered in Minnesota, United States), Johnson & Johnson (in New Jersey, United States), GE Healthcare (in Illinois, United States), and Royal Philips (in Netherlands). This study contributes to existing research in the field of outsourcing by providing a link between a firm's outsourcing strategy and its sustainable competitiveness from the view of the outsourcer's dynamic capabilities coping with the environmental change. A DCV-based research framework interprets the dynamic process of outsourcing resources and capabilities between the outsourcer and the outsourcee, and further investigates the role of successful outsourcing on the sustainable competitiveness of the established firms in technology-intensive industry. Moreover, the in-depth case study of multiple established firms nourishes our understanding of different outsourcing types in technology-intensive firms, which are classified by the characteristics of outsourced resource, and its alignment with resources owned by the outsourcer. The outsourcer's dynamic capabilities for

successful outsourcing, consequent benefits, and significance of the outsourcee's size according to each outsourcing type are also provided.

The remainder of this paper is organized as follows. Section 2 provides a theoretical background, while Section 3 describes the research method of this study. Section 4 develops the research framework from a review of relevant literature, while Section 5 analyzes the details of the case study, discussing the research results. Finally, Section 6 provides theoretical and managerial implications of the study and concludes with the suggested future work.

2. Theoretical Background

2.1. Tranditional Approaches to Outsourcing

The term "outsourcing" comes from the American terminology "outside resourcing", which means to get resources from the outside [24]. It is defined as an agreement in which a firm contracts-out a part of its internal process to another firm [25]. Outsourcing also refers to the reliance on external sources for manufacturing components and other value-adding activities [26]. It has been referred to as "disintegration of production", "one subtype of distributed work", or "vertical fragmentations" [27–29]. Traditionally, scholars considered outsourcing as "make or buy" decisions on intermediate goods to the use of temporary external services, or the process of entrusting non-core activities from internal manufacturing to an external firm specialized in particular operation [30,31]. Recently, the strategic importance of outsourcing, such as generating innovation or sustaining competitiveness, is gaining popularity [8,32].

Although the outsourcing concept has been proposed in various contexts, the theoretical foundation for explaining motivation of outsourcing and building a successful outsourcing relationship is mainly based on two streams, the economic perspective of TCE, and the capability sourcing perspective of RBV [33]. TCE, originally suggested by Williamson (1975) and built on the prior study of Coase (1937), has emerged as a predominant theory, explaining outsourcing activities via an interdisciplinary approach that combines economics with organizational theory and contract law [7,13,34]. The approach provides a definite guideline of when the firm will perform transaction in-house, and when it will turn to the external market [13,34,35]. TCE is concerned with uncertainty factors in deciding appropriate governance mechanisms that include competitive pricing and the necessity of specialized investments [36,37]. Cost reduction can be achieved through variable structural mechanisms, including the rearrangement of bureaucratic organization [12]. TCE posits that managerial choice between different governance structures, such as vertical integration or disintegration, i.e., contracting out, should be made to minimize the sum of transaction and production costs [5,13,14,34]. The basic premise of the perspective is that if the transaction costs of contracting out to the outsource outweigh the production cost advantages, firm managements should choose in-house production, rather than outsourcing [37,38].

Since a strategy focused on the cost reduction or efficiency is a part of complicated outsourcing phenomena, and therefore does not guarantee the long-term success or sustainable competitiveness of an organization; outsourcing is considered as a repetitive strategic decision on which business processes are to be developed by suppliers with superior resources or capabilities, aiming at the pursuit of sustainable competitive advantage [39,40]. Many technology-intensive firms acquire and integrate resources by means of strategic alliances with other firms, rather than by extending corporate boundaries [3,41]. In this respect, resource and capability-based theories can interpret outsourcing strategy and related organizational behaviors in a more proper way [19].

RBV, which considers a firm's valuable, rare, costly to imitate, and non-substitutable resources as a basis of its competitiveness, is another powerful theoretical approach to explain outsourcing relations [2,5]. Unlike traditional economic assumptions of TCE, RBV scholars propose that resources distributed across firms are heterogeneous and imperfectly mobile [2,34]. RBV argues that differences in a firm's resource endowment can result in differences in performance, assuming resource heterogeneity and immobility [2,42]. Penrose (2009) noted that a firm can be defined as a collection of productive resources, not merely an administrative unit [43]. As each firm has different resource assortments, it

searches for valuable and complementary resources to access those that it does not own [44]. RBV posits that through outsourcing activities, firms can maximize their value by building successful relationships with partners, exploring new markets, and gaining strategic market access, not merely minimizing costs [17,45]. This approach provides key issues in the outsourcing strategies, such as which types of resources can lead to high profits, which resources should be internally developed, and which resources should be acquired from external providers [46,47].

2.2. Dynamic Capability View of Outsourcing

Despite the importance of RBV in the field of outsourcing and strategic management research, the approach has been criticized for its static nature, especially in a volatile market, and its failure to clarify how resources are transferred and lead to sustainable competitive advantage [33,42,48,49]. It is necessary for firm managements to keep reconfiguring the boundaries of the core competencies to maximize value creation along with environmental change [7]. Hence, scholars have focused on dynamic capabilities as the forefront of a sustainable competitive advantage of a firm, along with the increase in the complexity and dynamism of the business environments. Whereas RBV focuses on the resource itself, DCV considers the processes of achieving new resource configurations in an integrated and coordinated way [50,51]. Dynamic capabilities transform operational-level capacities to enhanced value-creating routine and thus is critical to long-term survival of an organization [51,52]. Dynamic capabilities are conceptualized as a learned and stable pattern of collective activity, through which an organization systematically generates and reconfigures its operational-level routines to attain increased efficiency and flexibility [53]. A unique combination of resources through coordinated processes generates hard-to-imitate outcomes, which provide sustainable competitive advantage for the firm [50]. Thus, a firm's long-term success highly depends upon the firm's ability to develop dynamic capabilities [54–56].

In a technology-intensive industry, the development of dynamic capabilities is particularly crucial in outsourcing strategy, because it is related to knowledge management routines such as learning, sharing, integration, and reconfiguration between the outsourcer and the outsourcee, which in return results in sustainable value addition to the outsourcing process [12,23,57]. Combining current capacities with new resources or technologies acquired by interactions with other firms is a core activity in the innovation process [21]. From outsourcing, a firm can access a broader range of advanced and specialized technology available to the organization that can provide a more competitive resource base. Hence, dynamic capabilities allow the firm to increase the capacity to create, extend, or modify its resource base, thereby creating future value [17].

The dynamic capability construct in the outsourcing process is a significant factor affecting the successful outsourcing partnership, and as a result, explaining successful outsourcing performance [5]. The dynamic capabilities of the participants transform the network into a strategic net, and the potential complementarities of resources owned by the outsourcer and the outsourcee can turn into a higher outsourcing performance [58]. The created value from outsourcing practices can also be increased through a trust-based relationship and the willingness to learn from each other [12,59].

3. Research Method

Based on the arguments in the previous section, this study uses DCV as the underlying research framework that outlines the successful outsourcing process of established firms and its contribution to sustainability. Firstly, to construct the framework, we reviewed abstracts of literature that are collected by keyword or subject search (including "outsourcing" or "dynamic capabilities") in the database of SCOPUS and the Web of Science, and in addition, by snowball sampling from the reference list of each document. Secondly, we selected 43 research documents from abstract review which study outsourcing phenomena of firms from the perspective of dynamic capabilities or knowledge-based capabilities. Key concepts of outsourcing activities from the perspective of dynamic capabilities are derived by investigating the selected documents, which constitute our research framework.

We validated and concretized the constructed research framework by multiple case analysis of medical device firms. A qualitative case study provides abundant insight into a long-term complicated phenomenon that is difficult to be clearly understood [60,61]. Therefore, an in-depth case study design is fitted to investigate outsourcing capabilities, strategies, and created values of multiple firms that have been interacting for many years. The cases are closely analyzed through content analysis for published media sources on the basis of the established research framework. Content analysis is a research methodology that can be used to make relevant and valid inferences from context [62]. Mass media, such as corporate websites, newspapers, and industrial magazines, were selected as major sources of this study, since they provide rich amounts of news and official interviews relevant to major outsourcing events and strategic activities of firms for the period of interest. Therefore, secondary sources over a specific period are suited for qualitative research [46,63]. Content analysis procedure starts from inducing the original intention from content and developing a quantitative interpretation [52]. After establishing the research outline, the analysis develops category schemes for research, and validates them. We abstracted key phrases from each of the collected texts and classified them into proper categories that are developed in the stage of conceptual study. The developed categories can be adjusted or created according to the case study results. We performed the process of qualitative coding for the articles via Weft QDA, a qualitative data analysis software package, as is briefly illustrated in Figure 1. Figure 2 outlines our research procedure, where the initial framework based on literature review and the case study results complement each other to build final research framework.

To investigate the case of firms that have been successful in value creation for years, we selected four established medical device firms, i.e., the outsourcers, that are ranked in the top 4 by annual sales in 2017 (Figure 3). Since these four firms also maintain the top four medical device firms during that period (Figure 4), we claim that these four established firms are truly good examples of sustainable medical device firms in the last 10 years. Outsourcing activities of the four selected firms provide us with useful evidence that lead to their sustainable performances. A total of 45 outsourcing partners that performed outsourcing business with these top four medical device firms, including three duplicated firms, were investigated throughout the case analysis. We collected useful content published in the period (2010–2018) from diverse media sources including medical, economic, IT magazines and websites of the firms. To maintain objectivity, we reviewed each outsourcing event from multiple sources that are accessible online.



Figure 1. Qualitative coding procedure in content analysis (using Weft QDA).







Figure 3. Top 20 medical device firms by order of annual sales (in 2017). *Source: Medical Product Outsourcing Magazine (https://www.mpo-mag.com/)*.



Figure 4. Annual sales of the top-4 medical device firms over 10 years. *Source: Medical Product Outsourcing Magazine (https://www.mpo-mag.com/)*.

4. Research Framework

The dynamic capabilities of outsourcing firms are vital elements of a successful outsourcing process, and resultantly create values that are critical to their sustainable competitiveness in a volatile

market. In this respect, resources of outsourcer-oursourcee (i.e., antecedents), the outsourcer's dynamic capabilities necessary for outsourcing (i.e., process), and created value from the outsourcing activities (i.e., outcomes) comprise our research framework. We derive key elements and propositions constituting the framework by reviewing research documents that we collected by a keyword search (outsourcing & dynamic capabilities) in SCOPUS and the Web of Science.

4.1. Resources: Antecedents

According to RBV, a firm's superiority in specific resources explains the different performances among firms in the same industry, and thereby can be an important source of sustainable competitiveness [64–66]. Resources of both the outsourcer and the outsourcee are basic determinants of outsourcing strategy, whose potential complementarities turn into greater outsourcing performance by the organizational capabilities of firms [59]. Firms are motivated to outsource when they are in need of resources, or when they possess valuable resources to share [19]. Some studies suggest that firms with a higher level of internal resources are more capable of exploiting external resources [67,68].

For strategic purposes, the outsourcers increasingly focus on outsourcing various forms of resources, including physical resources, human resources, R&D knowledge, tacit know-how, and managerial systems [17,69]. Non-core resources that do not belong to the outsourcer's core competence, clear definition of which is necessary before the outsourcing decision, can be outsourced, in order to cut cost, or increase efficiency by better focusing on core technology [13,70]. Some studies suggest that firms should not outsource core competencies or related strategies to outsourcing partners, in line with the RBV or core competence theory [20,71,72]. Thus, the alliance structure between the outsourcer and the outsourcee can be affected by the characteristics of resources, e.g., core, or non-core characteristics, and alignment of the outsourcing resources, for example, supplementary (i.e., reinforcing resource base supplemented by similar resources) or complementary (i.e., expanding resource base complemented with dissimilar resources) [3,19]. We hereby posit the following research propositions:

Proposition 1. *Characteristics of resources owned by the outsourcer and the outsourcee, and their alignment affect outsourcing strategy.*

4.2. Dynamic Capabilities: Process

The outsourcer's dynamic capability to recognize the value of external resource or knowledge is a prerequisite in the process for successful outsourcing [73]. Environmental sensing capability is a capability that can be used to detect an opportunity or threat by exploring new technology and knowledge, thereby taking the appropriate measure for the particular situation [74–77]. This sort of cognitive flexibility is more critical as environmental uncertainty increases [46,78]. In short, the outsourcer needs a capability of sensing opportunities that recognizes customer's needs, as well as industrial technology trend, and searches for external resources to complement the current resource base of the firm.

In addition, the lack of internal firm resources can be compensated by focused learning from the appropriate outsourcee, if otherwise, it can never achieve on its own [79]. The outsourcer's absorptive capability to acquire and assimilate new recognized external resources for the purpose of applying it to commercialized ends is another driving force for a successful outsourcing relationship [73,77]. Innovation through adoption, i.e., the ability to draw ideas or knowledge from elsewhere, and build on them to meet the demand for changes in the start-up stage, becomes a powerful dynamic capability for a firm's sustainable growth [22,57]. Absorbing external technology from diverse sources to help maximize the benefits of a firm can nourish the development of operating routines and can enhance dynamic capabilities [53]. Absorptive capabilities have positive relationship with outsourcing partnership performance, in that the more both parties are willing to learn from each other, the higher the outsourcing performance [59]. Therefore, the following propositions are formulated:

Proposition 2a. The outsourcer's sensing capability contributes to a successful outsourcing process.

Proposition 2b. The outsourcer's absorptive capability contributes to a successful outsourcing process.

Dynamic capabilities enhance knowledge management practices of integrating absorbed foreign resources with internal capacities along the value chain system to create maximum values [23,80]. Integrated technology within a firm through interaction with the other organizations and the environment will contribute to launching the timely-developed products quickly to grasp the changes in business opportunities [22]. In large firms, system integrating capabilities to bundle diverse development fields, e.g., hardware, software, design, and manufacturing, in order to propose new solutions to market, are in the spotlight of management [68,81]. Thus, integrating capabilities to positively interact among the different fields of resources by transforming them into comprehensive sets of organizational skills that can create value in line with environmental changes are requisite dynamic capabilities for outsourcing [82].

Likewise, outsourcing value can be created through transferring technology that is internally accumulated or acquired from elsewhere [57]. According to market conditions or for strategic purposes, the outsourcer should be willing to share their knowledge with external vendors [79]. Dispersion of innovation activities through the value chains from outsourcer to key suppliers, i.e., outsourcee, can lead to opportunity seizing, such as exploring new markets, and timely launching of new products. In return, the outsourcee makes advances in innovation capability as a result of the outsourcing process [83]. Also, a firm's governance development to assist the technology flow, while protecting its core competences or intellectual properties from misappropriation, constitute important dynamic capabilities in many industrial sectors [77]. The following propositions emerge from the above arguments:

Proposition 2c. The outsourcer's integrating capability contributes to a successful outsourcing process.

Proposition 2d. The outsourcer's technology transfer capability contributes to a successful outsourcing process.

4.3. Value Creation: Outcomes

According to DCV, the routine of absorbing and sharing valuable resources among business partners improves an organization's strategic flexibility to changes in market or customer needs [50]. Flexibility is referred to as the organization's capability to meet an increasing variety of customer needs and spreads risks by diversifying business models or pioneering niche markets, which therefore is a core success factor in a volatile environment [1,84]. The advantage of managing properties that are owned by other independent suppliers is that the buyer can increase its flexibility, switching orders between suppliers as the market environment changes [85–87]. Many established firms adopt a simultaneous twin process of vertical integration and disintegration (or outsourcing) through the product life cycle, in order to maintain strategic flexibility [81]. The existing resource base of the firms can be reconfigured, e.g., acquired, changed, or discarded, to adapt them to the changing situation as a result of outsourcing activities [88]. Increased flexibilities of the firms help them spread risk by diversifying their business model and market in the modern volatile market. We posit the following proposition:

Proposition 3a. The firm's outsourcing activities help increase its strategic flexibility, diversifying its business model.

Many firms choose to outsource their non-core activities to reduce cost and increase efficiency [6]. As in the transaction cost perspective, the outsourcer can minimize the cost of product development and production by converting fixed costs into variable ones, reducing operating costs, reducing employees, eliminating unprofitable projects, and avoiding new infrastructure investments [89,90]. When a firm outsources non-core resources (e.g., maintenance, repair, or transportation), cost reduction is a primary

benefit from outsourcing, rather than capability enhancement [33,89]. For example, the outsourcer can reduce production cost by offshore production in a low wage country [5].

In addition to cost efficiency in the non-core activities, a firm's core activities of R & D become more efficient by outsourcing activities that are less related to their core competences. The outsourcer can invest more time and money in the area of strategic importance by hiving off non-core operations [91]. Outsourcers also have benefits from their partners by sharing the costs of R & D and new product launch in the global marketplace [92]. We thus propose as follows:

Proposition 3b. *The firm's outsourcing activities increase its cost efficiency in core activities, as well as non-core activities.*

Increased efficiency in the firm's outsourcing activities helps improve its agility. Agility can be referred to as a dynamic capability to respond quickly to customers' changing demands by making timely decisions and shortening production cycles [93]. The time-to-market launch of a firm's new products ahead of its competitors by reducing the R & D cycle is an important source of its competitive advantage [94,95]. Such a time-to-market capability becomes especially important in the market of short product life cycles [95]. Outsourcers will have benefits of enhanced time-to-market capability from successful outsourcing strategy. For example, an outsourcer could benefit its agility from reduced development time through R & D collaboration with the outsourcee, or from absorbing state-of-the-art technology that is already developed by the technology-provider [58,89]. The outsourcer can spare time to develop new products by focused learning from appropriate partners, rather than taking the risk of experimenting with new technology and developing all requisites in-house [79]. By outsourcing structures and organizational inertia [85]. Thus, we make another proposition, as follows:

Proposition 3c. The firm's outsourcing activities improve its agility to promptly grab emergent opportunities.

One of the core activities for the firm's growth is innovation, a central part of which involves acquiring valuable technology through outsourcing [7,96]. Ultimate value created by the firm's successful outsourcing process lies in the market-oriented innovation that puts a firm in a sustainable leadership position [8]. The firm can better achieve innovation in new product development, whether incremental or radical, through the outsourcing process of learning, integrating, and sharing cutting-edge technologies [86,95]. They can keep their own capacities up-to-date by managing and coordinating a network of best capabilities through outsourcing [6]. The more an outsourcer's organization is open to the external partners that can be potential sources of innovation, the easier such innovations can be brought [73]. As successful outsourcing is based on the constant identification of market needs, created value from the outsourcing experience includes market-oriented innovation that can attract consumers, and thereby dominate the market. We therefore derive our final proposition as follows:

Proposition 3d. The firm's outsourcing activities help achieve market-oriented innovation.

4.4. Framework Building

From the arguments and propositions described in the previous section, we propose a research framework as in Figure 5, which interprets the elements of a successful outsourcing process. Resources of the outsourcer and the outsourcee, whether core or non-core characteristics, are basic elements that motivate the outsourcing process. The outsourcer's dynamic capabilities to sense good partners along with technology trend or market needs, to acquire necessary resources (or technologies) and integrating them with internal capacities, and to transfer its own resources to the appropriate outsourcee for the purpose of maximizing benefits, form the basis of a successful outsourcing process. Through the successful outsourcing experience, the outsourcer can create greater value for consumers and

enhance its dynamic capabilities, in the form of improved strategic flexibility, agility in market response, increased efficiency, and market-oriented innovation. Moreover, an outsourcer can take advantage of international networks that are expanded through outsourcing experiences, which can become a bridge to a broader network of foreign partners [85,97]. Such outcomes of successful outsourcing process create a virtuous cycle of sustainable competitiveness through accumulated capabilities.



Figure 5. Research framework of outsourcing process: virtuous cycle of sustainability.

5. Case Analysis

5.1. Case Description

In the medical device industry, which is technology-intensive and rapidly growing, the competition among firms has recently become intense and diversified. New entrants to the industry are gaining competitiveness by effective ways, such as outsourcing necessary resources, or acquiring businesses. The reason for the high growth of the medical device market is analyzed from various viewpoints, such as population aging, and the growth of emerging countries. According to World Health Organization's report, medical supply and expenses for the elderly population are likely to be a great burden on society, because cardiovascular, cerebrovascular, or respiratory diseases that mainly affect them are increasing [98]. In addition, expansion of the middle class and rising GDP in China, India, Latin America and other emerging countries also drive demand for medical devices and services [99]. Nevertheless, the number of medical professionals and facilities currently available is not sufficient, and that is why there is a rising demand for advanced medical devices that are capable of early diagnosis and minimally invasive treatment, and IT convergence products for home care and self-care.

Established firms in the medical device industry are using outsourcing strategies in a variety of business activities, such as R & D, manufacturing, service, and licensing. The outsourcing market in the medical device industry is expected to grow by an average of 10.5% annually by 2025 [100]. Although outsourcing is a major activity in the medical device industry and is highly significant from the perspective of corporate strategy, there is still a lack of relevant academic research. Our case study includes the top four established firms successful in the medical device business—Medtronic, Johnson & Johnson (J & J), Royal Philips, and GE Healthcare—for detailed investigation of the interrelationship among their resources, capabilities, strategies, and created values in the outsourcing activities from 2010 to 2018.

5.2. Research Results

5.2.1. Evidence from the Case: Resources and Outsourcing Types

As a result of the content analysis for the four leading, established firms (i.e., outsourcers) in the medical device industry, we go into detail of the research framework in Figure 5. Firstly, we found four distinctive types of outsourcing events according to the characteristics of outsourced resources, i.e., core or non-core, and their alignment with resources owned by the outsourcer, i.e., supplementary or complementary. In the outsourcing events of type I and II, the firms outsourced core resources for the purpose of supplementing or complementing their current capabilities, respectively. While in type I outsourcing, mutually complementary resources of the outsourcer and the outsourcer acquired resources that could reinforce its capabilities by supplementing its resource base. In type III, the firm outsourced its non-core activities, such as production and payroll service, which as a result, supplemented their core activities. In type IV, the outsourcer transferred a core part of its resource base to the outsourcing partner, for the purpose of taking advantage of the outsourcee's local network and expanding the marketplace. Figure 6 indicates the four types of outsourcing, along with the characteristics and alignment of the outsourcing resources.

Several items of evidence from the case according to each outsourcing type are described as follows. The Appendix A briefly summarizes the main outsourcing events with key phrases from the articles, media sources and classified outsourcing types.



Resource alignment

Figure 6. Four outsourcing types according to the resource characteristics and alignment.

• Type I: Outsourcer (Core resource) + Outsourcee (Core resource)

Collaboration in technology with a variety of partners has been the most common type of outsourcing in the technology-intensive firms engaged in the medical device business. The outsourcer develops diverse innovative products that are well adapted to the changing market, by integrating appropriate external technologies with internal competence, thereby creating synergy. In this type of outsourcing, core resources of both the outsourcer and outsourcee, which are dissimilar and mutually complementary, are involved in outsourcing activities. Many events summarized in the Appendix A (ID: M1–M6, P1–P10, G1–G8, J1–J4) support that the four established firms have combined their strengths with appropriate partners to expand their capabilities and diversify business models, several items of the evidence of which are described below.

Medtronic has integrated its sophisticated medical technologies with another core technologies such as Fitbit's automatic activity tracking technology, which provides insights into how exercise affects glucose levels for more effective diabetes care management (ID: M1), and IBM's artificial intelligence unit, which enabled the diabetes devices of Medtronic to predict whether a diabetic person will have

low glucose (ID: M2). Medtronic also took advantage of its market-leadership in diabetes management in collaboration with Qualcomm Life's expertise to create more affordable, easier-to-use Continuous Glucose Monitoring (CGM) systems that can provide near real-time and retrospective glucose data (ID: M4). Medtronic's range of future solutions, jointly developed with Samsung Electronics, could improve the way people with diabetes live, by remotely viewing diabetes data and ultimately integrating mobile devices into diabetes management systems (ID: M5).

In the case of Philips, a partnership with Amazon connected millions of devices to the Internet of Things (IoT) using Amazon Web Services (AWS) (ID: P1). Philips also made a strategic alliance with Salesforce.com, leveraging Philips' leadership in medical technology and Salesforce.com's leading position in enterprise cloud computing by delivering a cloud-based healthcare platform (ID: P3). For the purpose of a long-term collaboration with partners, Philips secured certified compatibility between its products and the outsourced systems (ID: P8).

GE Healthcare also established its cloud system that integrates powerful IT solutions of large-sized system integrators and leading-edge independent software vendors across a range of care areas (ID: G5, G6).

J & J integrated its core capabilities with experts in 3D-printing technology in order to develop 3D printed products in the areas of orthopedics, eye health, and consumer health. For example, 3D printed surgical devices increase the predictability and efficiency of surgery, and pharmaceutical researchers can recognize how the drug compounds will react in a real human body by measuring the toxicity levels of compounds on 3D printed kidney cells or liver cells (ID: J1–J3).

• Type II: Outsourcer (Core resource) → Outsourcee (Core resource)

When the firm plans to catch up with cutting edge technology of the industry to fulfill customer needs promptly, it particularly needs external partners with advanced technologies to fill the technology gap. In this type of outsourcing, the outsourcer acquires the necessary external resources and sometimes the relevant organization or company. The firm would also make a cross license agreement with the technology provider to use the necessary intellectual property, in the case that mutual interest is met. The outsourced resources, which are core resources of the outsourcee in this type, are supplementary to resources owned by the outsourcer, thereby enhancing the core competence of the outsourcer. Part of the events in Appendix A (ID: M7, P11–P14, G9–G11) correspond to this type.

Philips could strengthen its innovation pipeline of catheter-based therapy devices by acquiring CardioProlific, because CardioProlific's technologies are supplementary to Philips' portfolio of image-guided therapy devices. The acquisition of CardioProlific underpins Philips' strategy to develop new solutions for clinicians to guide and confirm the adequate therapy for cardiac and peripheral vascular disease (ID: P11). In addition, Philips has signed an agreement to acquire EPD Solutions, an innovator in image-guided procedures for heart rhythm disorders, which supplements Philips' portfolio of interventional imaging systems, smart catheters, planning and navigation software, and services. As a result of the acquisition, Philips could introduce new solutions in the fast-growing EUR 2+ billion market for the image-guided treatment of cardiac arrhythmias (ID: P12). Philips also acquired Remote Diagnostic Technologies (RDT), a leading innovator of advanced solutions for the pre-hospital market, providing cardiac therapy and data management, for the purpose of enhancing Philips' Therapeutic Care business and strengthening its leading position in the EUR 1.4 billion resuscitation and emergency care market (ID: P14).

In the case of GE Healthcare, the firm reinforced its capability to provide a variety of ultrasound disinfection solutions to customers by outsourcing CS Medical's products, which provides healthcare professionals the tools to properly reprocess the delicate ultrasound probe, thereby minimizing healthcare-acquired infections (ID: G9). GE Healthcare also outsourced SonoSim's solution that offers training from the basics to hands-on simulations of real-world, pathologic cases from actual patients. GE Healthcare invested \$13 million to help build a modernized healthcare system in Kenya by building

the GE Healthcare Skills and Training Institute, for the purpose of training healthcare professionals in the use of the up-to-date technology (ID: G11).

• Type III: Outsourcer (Core resource) ← Outsourcee (Non-core resource)

The management may decide to outsource if the sum of production and transaction cost is less than the internal production cost. This strategy, where the firm outsources a non-core part of its operation, for example, offshoring to a low-wage country, is the most common strategy used in the traditional manufacturing firms for the purpose of minimizing overall cost. Although not many outsourcing events of medical device firms belong to this type, several events in Appendix A (ID: M8, G12, J5, J6) could be classified to this type.

For the purpose of R & D efficiency and cost reduction, Medtronic outsourced transactional and repetitive jobs that do not require high-level knowledge of Medtronic. For example, the firm hired the well-known outsourcing and IT services firm Cognizant, whose department provided customer service and phone-answering functions for Medtronic (ID: M8).

GE Healthcare also outsourced the non-core process of training doctors and nurses on all the machines General Electric manufactures, from MRI machines to ultrasound equipment. Such operation is typically cumbersome and expensive, involving flying doctors to training sites, or else sending experts out to hospitals. However, a new partnership with InTouch Health helped to bring remote education, training and technical support for a bunch of GE imaging products (ID: G12)

J & J has signed an exclusive original equipment manufacturer (OEM) agreement with Nova Biomedical to provide its point-of-care blood glucose testing systems to hospital customers in China. By outsourcing non-core process to the emerging market, J & J could serve the changing needs of a fast-growing market with increased efficiency (ID: J5). Besides, J&J could save time in reconciling payroll issues, and focus more on its core activities, by partnering with PayrollServe, a specialized firm in payroll matters (ID: J6).

• Type IV: Outsourcer (Core resource) + Outsourcee (Non-core resource)

Another type of outsourcing classified from the case analysis occurs when the outsourcer plans to expand the business by exploring new markets. In this type, the outsourcer transfers resources to the outsourcee that typically belongs to a developing country with a relatively low level of industrial technology. For example, the outsourcer grants the local partner a license to develop the products, or shares the relevant know-how with the firm, whereas the local partner can acquire core technology from the outsourcer, thereby enhancing its own competence. Such outsourcing can be supported by the local government, because the collaboration may bring a positive ripple effect to the entire national industry. Even though the outsourcer does not receive an immediate return from the transaction, it could gain access to a new niche market, and from a long-term perspective, take the initiative in the industrial ecology by securing more and more distribution networks. Part of the events in Appendix A (ID: M9, P15–P18) exemplify this type of outsourcing.

As an emerging market strategy, Philips has established a unique cooperation format with Russian firms of leading industrial networks in the local market. The firm partnered with Electron, a leading Russian medical equipment manufacturer, supporting the modernization of the healthcare system at technology and even strategy levels, by bringing in Philips' global expertise to serve the needs of the Russian healthcare system (ID: P15). Philips also cooperated with MEDSI, a leading provider of patient-centered quality care in Russia, taking advantage of the healthcare facilities of MEDSI. This new partnership provides MEDSI with guaranteed affordable access to the Philips' latest healthcare equipment, technology, and international expertise in hospital management, beyond the traditional supply of equipment (ID: P16). In addition, with the signing of a ten-year strategic partnership agreement with the Expert Group of Companies, Russia's leading networks of healthcare centers and clinics, Philips provided its clinical expertise, consulting services, and technology planning for multidisciplinary medical centers (ID: P17).

In the case of Medtronic, the firm has developed an innovative partnership with the National Institute of Hospital Administration, an influential think tank under China's National Health, to carry out research projects focusing on building an integrated care system for patients with diabetes. This partnership shows an effort to collaborate with an important organization to access data necessary to understand the clinical and economic value of therapies, and how they can best be utilized in a patient population. In this joint endeavor, Medtronic shared with the Chinese government the extensive experience it has accumulated through dealing with a similar mechanism in other countries around the world (ID: M9).

5.2.2. Dynamic Capabilities and Value Creation

The case study results also explain details of other elements of the proposed research framework: dynamic capabilities and value creation. The four leading, established firms showed different dynamic capabilities in the events of different outsourcing types. In all types of outsourcing, the sensing capability of understanding market demands and continuously exploring an appropriate partner with superior or cheaper resources play a key role in the successful outsourcing process. For example, Omar Ishrak, chief executive officer of Medtronic, recognized market needs and realized where to focus, by saying, "Diabetes is unfortunately rising in prevalence around the world, driving up system costs and, ... we believe there is tremendous opportunity to better align care across the diabetes care continuum through new technologies and patient care management strategies" (ID: M6).

Specifically, the outsourcer that plans type I outsourcing requires integrating capability to combine different kinds of resources based on mutual trust and cooperation among the outsourcing partners [10,54]. When GE Healthcare collaborated with Zenith Technologies, it announced that the two companies integrated process control systems, and other supporting technologies with GE Healthcare's start-to-finish technologies for the biopharmaceutical manufacturing industry (ID: G4). As a result, the outsourcer could create market-oriented innovation. "Using the Arterys System together with cloud computation, the diagnoses can be made in a non-invasive MR imaging procedure that is quick, accessible and easy for both the patient and the radiologist," said Fabien Beckers, Arterys CEO, in a statement in 2016 (ID: G1). Also, the partnership offered more choices to customers by diversifying the solutions. GE Healthcare announced that it will bring IT solutions for clinicians across a range of care areas, by collaborating with two large systems integrators and seven leading-edge software vendors (ID: G5). In some cases, type I outsourcing brings benefits of lower total costs, as announced by Jim Davis, GE Healthcare's management, in 2011. "Research in this area aims to bring benefits of lower total costs, ... and makes this an attractive opportunity for collaboration" (ID: G8).

In type II outsourcing, the outsourcer needs absorptive capability to acquire, learn, and share external knowledge within the organization. Through the outsourcer's absorptive capability, the absorbed external resources can form a knowledge evolution cycle of promoting the outsourcer's core competence and exploring more advanced technologies. In particular, established firms need a culture of openness to successfully acquire external knowledge, while overcoming organizational inertia, and accepting unfamiliar changes. As a result of this outsourced up-to-date technologies in a shorter period of time than developing entire technologies internally. Strategic flexibility can be increased by diversifying products from utilizing multiple outsourcing partners, and thereby spreading risk in a volatile environment. Several announcements from the management show evidence of the value created by the outsourcing, such as the following statements: "… this agreement reinforces GE's ongoing pledge to provide a variety of ultrasound disinfection solutions to help our customers improve patient outcomes" (ID: G9), "Medtronic acquired the OsteoCool technology and … the OsteoCool System expands our Pain Therapies portfolio" (ID: M7).

In type III outsourcing, a sensing capability of the decision maker to continuously relocate resources wherever they are needed to minimize time and cost in accordance with the changing market condition is a crucial element. Although this type of outsourcing is not directly relevant to long-term

value creation, the outsourcer can focus internal capacities on its core business area by outsourcing the non-core part of business to reduce time and cost, thereby increasing R & D efficiency. Several articles support the kind of values the outsourcing activities have created, as an official Medtronic spokesperson said, "... the job cuts will allow Medtronic to be more adaptive to fluctuations in market demands and maintaining service levels to our customers" (ID: M8). Also, Business Journal (2016) reported, "... industry spends thousands of dollars annually paying for travel and accommodations to train physicians on their devices ... collaboration with InTouch Health enables network connectivity, hands on interaction, and an intuitive user experience" (ID: G12). In addition, the outsourcer can speed up time to market by utilizing production facilities or market accessibility of the outsourcee, as Prnewswire (2014) reported, "the business has signed an OEM agreement with Nova Biomedical Corporation to market ... testing systems to hospital customers in China" (ID: J5).

Lastly, the outsourcer in type IV should be capable of sharing the core technology with the partner, whilst maintaining control over the relationship. Transferring parts of business processes or resources that are a core or complementary-to-core competence of the outsourcer could be risky, if the outsourcing partner spreads the acquired know-how to a third party and uses it to compete with the outsourcer. For this reason, the outsourcer should evaluate the detailed strategic value of the transferring resources, as well as the estimated benefits from the outsourcing, maintaining close contact with the partner. Evidence from Philips shows the expected benefits from this type of outsourcing. As an emerging market strategy, Philips explored the Russian healthcare market, utilizing the broad network of local Russian partners and infrastructure in the region, in return for supporting Russian healthcare industry by providing core resources to the local firm. Philips has improved the local healthcare system by developing products specialized in Russian under joint branding with the local partner. Ronald de Jong, CEO of Philips Healthcare said, "This unique partnership with Electron is a next step in executing on our emerging markets strategy for our healthcare sector, ... will support the modernization of the healthcare system at technology, economy and even strategy levels by bringing in Philips' global expertise to serve the real needs of the Russian healthcare system" (ID: P15). Such outsourcing strategy increased the firm's strategic flexibility, by expanding its business to local and global markets.

To summarize this section, Figure 7 illustrates the outsourcer's key dynamic capabilities that are required for successful outsourcing and created value according to the outsourcing types. Also, Table 1 briefly provides our research results supported by the case study, for each proposition suggested in Section 4.



Figure 7. Cont.



Figure 7. Outsourcer's dynamic capabilities and created value according to the outsourcing type.

Proposition	Research Results Supported by the Case Study
1	<i>Characteristics of resources owned by the outsourcer and the outsourcee, and their alignment affect outsourcing strategy.</i>
2a	The outsourcer's sensing capability positively affects "type I–IV" outsourcing.
2b	The outsourcer's absorptive capability positively affects "type II" outsourcing.
2c	The outsourcer's integrating capability positively affects "type I" outsourcing.
2d	The outsourcer's technology transfer capability positively affects "type IV" outsourcing.
3a	<i>The firm's "type I, II, III, IV" outsourcing activities help increase its strategic flexibility, diversifying its business model.</i>
3b	<i>The firm's "type I, II, III" outsourcing activities increase its efficiency in core activities as well as non-core activities.</i>
3c	<i>The firm's "type I, II, III" outsourcing activities improve its agility to promptly grab emergent opportunities.</i>
3d	The firm's "type I, II, IV" outsourcing activities help achieve market-oriented innovation.

Table 1. Research results for the suggested propositions.

5.2.3. Firm Sizes and Outsourcing Types

We further analyzed the frequency of the outsourcing events for each outsourcing type in different firms. From the content analysis of the four large established firms in medical device business, type I outsourcing, characterized by the integration of mutually complementary resources, appeared the most, while type III, i.e., events of outsourcing non-core activities, was the least as indicated in Figure 8. Compared to the capital-intensive industry where an outsourcer commonly benefits from economies of scale by outsourcing non-core operations to lower-cost foreign firms, our research case of technology-intensive firms shows much more outsourcing events related to technology or capability sourcing than cost reducing activities.

In addition, we investigated the significance of firm size by the outsourcing type. Large-sized enterprises (LEs) in the medical device industry are usually equipped with their own manufacturing facilities in addition to their inherent capabilities, while small and medium-sized enterprises (SMEs) utilize their strong points in technology or service in partnering with LEs. Since the outsourcers considered in this case study are LEs, the 45 outsourcees (including three duplicate firms) were classified as 18 SMEs and 27 LEs for further analysis. Although there are several different criteria of classifying LEs and SMEs, this study defined the firms with less than 500 employees as SMEs, according to the OECD's recommendation in 2005 [101–103].

Figure 9 indicates the frequency of outsourcing events for each outsourcing type by the outsourcee's size, i.e., "LEs vs. LEs", and "LEs vs. SMEs", because this study focuses on four large-sized outsourcers. The result shows that the outsourcing case in type I is not biased to either LEs or SMEs. The four

established firms partnered with the firms that own valuable, indispensable technologies, regardless of their size. In type II, the outsourcers partnered more with SMEs that are advanced in the required technologies in the form of technology or business acquisition at a reasonable cost, which accelerated business growth of the outsourcers. As a special case in type II, Philips made a deal with a large firm, Masimo, to resolve a patent dispute between them. As a result of a verdict in 2014, Philips payed \$466 million to use Masimo's patented technology that was indispensable for Philips to innovate and move forward.

We also found that the outsourcing events in type III and IV were biased to "LEs vs. LEs". In three cases of type III outsourcing, the outsourcers partnered with LEs that could provide manufacturing operation or payroll services promptly at a reduced cost, which makes the non-core part of business more efficient. In the other case of a partnership with a SME, InTouch, the outsourcee provided training solution to GE Healthcare, reducing time and cost remarkably. In the case of type IV, all five cases corresponding to the type were in the form of "LEs vs. LEs". In this type, the outsourcers could efficiently penetrate the emerging market through the influential local LEs, utilizing the broad networks and established regional facilities of the local firms. Table 2 briefly summarizes the outsourced resources in "LEs–LEs" and "LEs–SMEs", according to the outsourcing types.



Figure 8. Frequency of outsourcing events by outsourcing types and different outsourcers.



Figure 9. Frequency of outsourcing events by outsourcing types and the outsourcee's size.

Туре	LEs-LEs	LEs-SMEs	
Ι	Valuable technologies complementary to t	s complementary to the outsourcer's core competence	
II	Superior technologies or services (Settlement of patent dispute)	Superior technologies or services	
III	Manufacturing facilities, payroll service	Training solution	
IV	Broad networks, facilities in local market	-	

Table 2. Outsourced resources in large-sized enterprises to large-sized enterprises ("LEs–LEs") and large-sized enterprises to small and medium-sized enterprises ("LEs–SMEs") according to the outsourcing types.

6. Implication and Future Research

6.1. Implication

This study focuses on the outsourcing activities of established firms as an essential process to sustain competitive advantage, particularly in a volatile, technology-intensive industry. The suggested research framework, including resources, dynamic capabilities, and resultant value creation, comprehensively interprets outsourcing phenomena from a long-term perspective, which are supported and concretized by the in-depth case study results. This study proposes that the four outsourcing types classified by the characteristics of outsourced resources, i.e., core or non-core, and its alignment with the existing resources of outsourcer, i.e., complementary or supplementary, require different dynamic capabilities of the outsourcer for successful outcomes. In addition, case study results support that successful outsourcing experience brings a variety of value, such as market-oriented innovation, strategic flexibility, agility, and efficiency, to the outsourcer, which in turn enhances the dynamic capabilities of the firm. According to previous studies on DCV, elements of dynamic capabilities are key enablers to sustain a firm's competitive advantage in volatile environments [46,50,77,92]. We hereby claim that outsourcing activities of the leading firms have promoted development of the organizational dynamic capabilities, consequently leading to sustainable competitiveness. The research results also indicate the different frequency of outsourcing events according to outsourcing type and the outsourcee's size. From the large-sized outsourcees, the outsourcers took advantage of the established facilities or broad networks, as well as complementary resources; while from the small and medium-sized outsourcees, they outsourced superior technologies or services. The proposed research framework based on DCV and supported by in-depth case study results can be utilized in the outsourcing and sustainability research in the diverse industrial fields.

Our study also provides several managerial implications for the decision-makers of established firms. Our study results suggest that top management must develop the organizational dynamic capabilities necessary for the selected outsourcing type, which were found to be significant for sustainable competitiveness of established firms. Sensing capability is a prerequisite for all types of outsourcing, as is shown in the case study results. Continuous customer-related surveying and participation in the regular technology exhibition of outsourcing partners help decision-makers strengthen their sensing skills and formulate a suitable outsourcing strategy. Absorptive and integrating capabilities are also critical dynamic capabilities for continued success through a successful outsourcing experience. Collaboration between the outsourcer and the outsourcee, which is based on mutual trust and communication, enhances the capability of integrating internal and external resources, and a culture of openness helps an organization overcome organizational inertia and absorb external technology more efficiently. In addition, technology transfer can be strategically important for exploring new markets, with utilizing the broad network of local partners and infrastructure in the region. When an outsourcer transfers its core resource to the outsourcee, as in the case of Type IV, the management should minimize the leakage of core competence by introducing contract clauses that clarify the restitution of core information at the end of the project.

We further propose that top management must accurately diagnose the strengths and weaknesses of the current organization, before deciding on a proper outsourcing strategy. The management needs to define which part of the business process forms core or non-core resource, and which resource should be outsourced. As a result, the management can decide which type of outsourcing can bring maximum value to the organization. In addition, top management must consider changing outsourcing strategy depending on situation, to sustain competitiveness in a dynamic environment. It could be necessary for the outsourcer to switch outsourcing partner, or shift from outsourcing to in-house development, and vice versa. However, as an excessive dependence on outsourcing partners can impair an outsourcer's capability to generate its own know-how optimized for a variety of situations, top management must always focus on enhancing the firm's own core competence from the accumulated experience of successful outsourcing.

6.2. Future Research Directions

We suggest future research directions that could not be covered by this study. Firstly, as this study focused on the capability and consequent benefits only from the perspective of the outsourcer, future research conducted from the perspective of the outsourcee could be helpful to better understand the interaction between the outsourcer and the outsourcee. Secondly, as this study investigated the single industry case, additional research on different technology-intensive industry, e.g., pharmaceutical or electronics industry, is suggested, to investigate the difference in outsourcing type or strategy according to industry. Finally, as the scope of this study could not cover individual outsourcing events and their consequences, negative effects of specific events could have been overlooked. We therefore suggest additional longitudinal study tracing specific outsourcing events for years, which interprets detailed causal relationships between each of the outsourcing events and the consequent performance. The suggested future research is expected to enrich the outcomes of this study and shed a light on the field of outsourcing and sustainability.

Author Contributions: Conceptualization, B.P.; methodology, J.K.; validation, B.P.; data curation, J.K. and J.P.; writing—original draft preparation, B.P.; writing—review and editing, H.L.; supervision, H.L.

Funding: This research received no external funding

Acknowledgments: This work was supported by the National Research Foundation of Korea (NRF) grant funded by the Korea government (MSIT) (No.2018R1D1A1B07050139).

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

This appendix summarizes part of the evidence described in empirical research findings.

Case of Medtronic

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
M1	Fitbit	The partnership brings Medtronic's medical technology together with the automatic activity tracking of Fitbit, which provides insights into how exercise affects glucose levels for more effective diabetes care management. The collaboration between healthcare and technology is more important than ever, to simplify daily diabetes management.	Medtronic (2016)	Ι
M2	IBM	Medtronic and International Business Machines Co. (IBM)'s Watson Health unit outlined a tool to predict whether a diabetic person will have low glucose by integrating IBM's artificial intelligence and Medtronic's diabetes devices. Medtronic has been partnering with various technology companies to layer in more analytics.	ZDNET (2019)	Ι

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
М3	Nutrino	"By working with Nutrino, we will enhance our customers' daily lives and will also help us collect insights to inform how we approach diabetes management in the future," said the president of Global Diabetes Service and Solutions, Medtronic Diabetes.	Nutrino (2016)	Ι
M4	Qualcomm	Medtronic's market-leadership in diabetes management could be leveraged with Qualcomm Life's expertise to create more affordable, easier-to-use Continuous Glucose Monitoring (CGM) systems, including a new sensor and smaller design that can provide near real-time and retrospective glucose data.	Medtronic (2016)	Ι
M5	Samsung	Medtronic and Samsung Electronics are combining their strengths to develop a variety of future solutions that will improve the way people with diabetes lives, from remotely viewing diabetes data to integrating wearable and mobile devices into diabetes management systems.	Medtronic (2015)	Ι
M6	Sanofi	According to a joint statement of the companies, Sanofi's extensive insulin portfolio and drug development expertise will be combined with Medtronic's technology in continuous glucose monitoring. "Medtronic is committed to taking a broader approach, expanding beyond our core strength, to deliver superior clinical outcomes at an affordable price," said the CEO of Medtronic.	Medtronic (2014)	I
М7	Baylis Medical	Medtronic acquired the OsteoCool technology and associated intellectual property from Baylis Medical, partnering with the company to further innovate the system. "The OsteoCool System of Baylis Medical expands our Pain Therapies portfolio," said Julie Foster, general manager of the Pain Therapies business at Medtronic.	Medtronic (2016)	П
M8	Cognizant	The affected positions are lower and operational-level jobs that Medtronic called repetitive or transactional in nature, which do not require high-level knowledge of Medtronic. Medtronic is hiring the well-known IT services firm Cognizant, whose department will provide customer service and phone-answering functions for Medtronic.	MDDI (2016)	III
M9	Institute of Hospital	Chris Lee, the President of Medtronic Greater China said: "In this joint endeavor to help China prepare a nationwide Type 1 diabetes integrated pathway, we will share with the Chinese government the extensive experience we have accumulated through dealing with a similar mechanism in other countries around the world."	Medtronic (2013)	IV

• Case of Philips

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
P1	Amazon	Philips and Amazon will connect millions of devices to the Internet of Things (IoT) using Amazon Web Services. The partnership is likely to extends the health data management connectivity and Philips' HealthSuite digital platform technology.	Medtronic (2015)	Ι
P2	Banyan	Philips' patient monitoring and handheld diagnostics technologies are united with Banyan Biomarkers' strength in biomolecules. "The joint development agreement with Banyan Biomarkers will help Philips to expand into this fast-growing market and become a leader in the new market."	Philips (2017)	Ι

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
P3	Salesforce	Royal Philips announced a strategic alliance with Salesforce.com to deliver a cloud-based healthcare platform, leveraging Philips' leadership in medical technology, and Salesforce.com's leading position in enterprise cloud computing. "Together with Salesforce, we have a tremendous opportunity to reshape the way healthcare is delivered and provide better access to data."	Salesforce (2014)	Ι
P4	Image Stream Medical	Philips signed an agreement with Image Stream Medical to further expand its solution to integrate its hybrid suite and interventional lab solutions with integrated video and live streaming technologies. Integrated solutions that enable physicians to optimally perform their tasks is necessary, as the trend towards image-guided minimally invasive therapies continues to grow.	Philips (2014)	Ι
P5	MIT	Philips announced it has teamed up with Massachusetts Institute of Technology (MIT) to research the use of Philips ultrasound technology and MIT physiological modeling as a less invasive way to measure intracranial pressure (ICP), to manage patients with brain injuries. "We were surprised by the breadth and richness of the competency at MIT and being extremely networked, they have a global vision," said van Houten, the global head of Philips Research.	MedDevice (2015)	Ι
P6	Pegasystems	Royal Philips, in collaboration with Pegasystems, enables care service providers to better personalize care and improve health outcomes. By connecting the Pega Care Management Application to the Philips HealthSuite Cloud platform, service providers can remotely manage care with connected health applications and provide immediate access to a patient's health status.	Market-wired (2016)	I
P7	Qualcomm	Philips' cloud-based platform integrating health data from a variety of devices joins Qualcomm Life's 2net Platform, providing HealthSuite users seamless access to connected medical devices, including medication dispensers, ventilators, and blood glucose meters.	Fierce-Biotech (2016)	I
P8	Hansen	Hansen Medical and Royal Philips secured certified compatibility between Philips' Allura interventional X-ray systems and Hansen Medical's Magellan Robotic System, as a milestone in their long-term collaboration. "We believe that robot-assisted interventional techniques have great potential."	Philips (2013)	Ι
Р9	Trice Imaging	"Philips is committed to providing innovative solutions for global customers, and our partnership with Trice enhances our capability to provide world class ultrasound solutions," said Sean Gallimore, VP of Global Marketing of Ultrasound for Philips.	Trice Imaging (2014)	Ι
P10	Vidyo	The Philips-Vidyo platform will uniquely satisfy an increasing need for easier access to high-quality telehealth. We are excited to collaborate with Vidyo to leverage its cutting-edge telepresence technologies to enhance the scalability of our telehealth solutions. Philips is committed to working with industry leaders to better address healthcare needs by expanding telehealth services," said Mike Mancuso.	Vidiyo (2012) MobiHealthNews (2012)	Ι
P11	CardioProlific	"The acquisition of CardioProlific will strengthen our innovation pipeline," said Bert van Meurs at Philips. As CardioProlific's technologies are supplementary to Philips' portfolio of image-guided therapy devices, the acquisition underpins Philips' plan to develop new solutions for clinicians to guide and confirm the appropriate therapy for cardiac and peripheral vascular disease.	MedDevice (2017)	П

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
P12	EPD solution	Philips has signed an agreement to acquire EPD Solutions, complementing Philips' portfolio of interventional imaging systems and services. The company introduced new solutions in fast-growing market for image-guided treatment of cardiac arrhythmias. "EPD's breakthrough innovation provides unique and detailed 3D anatomical information of the heart," said CEO of Royal Philips.	Philips (2018)	п

• Case of Philips (Cont'd)

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
P13	Masimo	The partnership includes joint marketing and sales programs in North America, Asia and Europe for Masimo's non-invasive sensor technologies in conjunction with Philips' patient monitoring and select therapy solutions.	Philips (2016)	п
P14	RDT	RDT's portfolio of comprehensive connected emergency care solutions complements Philips' Therapeutic Care business. RDT will also strengthen Philips' ability to drive digital transformation in healthcare and explore new markets.	MedDevice (2016)	п
P15	Electron	"This unique partnership with Electron is a next step in executing on our Emerging Markets strategy," Ronald de Jong, CEO for Philips Healthcare said. "Philips will support the modernization of the healthcare system at technology, and even strategy levels by bringing in Philips' global expertise to serve the needs of the Russian healthcare system."	Euronext (2010)	IV
P16	MEDSI	This new partnership provides MEDSI with guaranteed affordable access to the latest healthcare equipment, technology, and international expertise of Philips, beyond the traditional supply of equipment.	Philips (2014)	IV
P17	Expert Group	Royal Philips has signed a 10-year strategic partnership agreement with the Expert Group of Companies, one of Russia's leading network of healthcare centers and clinics. Philips will provide its deep clinical expertise, consulting services and technology planning.	Philips (2014)	IV
P18	ROSATOM	Philips said it plans to license its knowledge for the production and maintenance of medical devices to Rosatom, and it will provide training to Russian engineers. "Partnerships between global and local firms are key enablers to improve local healthcare systems," said Steve Rusckowski, CEO of Philips Healthcare.	Euronext (2014)	IV

• Case of Johnson & Johnson

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
J1	HP	Under the partnership, Johnson and Johnson (J & J) and Hewlett Packard (HP) will pool their scientific, clinical, material science and technological expertise, to develop products that can be manufactured quickly and customized to the needs of individual patients. Experts from both companies will work with long-term plans to make 3D-printed products in the areas of orthopedics.	Medtronic (2016)	Ι

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
J2	Carbon 3D	J & J has picked Carbon 3D to develop custom surgical devices. Unlike other technologies that may 3D print these devices in a matter of hours, Carbon 3D will be able to produce them in a matter of minutes.	3dprinting-indi (2016)	ustry I
J3	Organovo	The companies will work together to evaluate the use of 3D bio-printed tissue in a drug discovery setting. By 3D printing tissues, pharmaceutical researchers can measure the toxicity levels of drug compounds on 3D printed liver cells, kidney cells, etc. This potentially reduces the need to test on animal subjects and can bring the drugs to market more quickly.	3dprint (2014)	Ι
J4	WellDoc	J & J's partnership with WellDoc, a startup developing mobile solutions for Type 2 diabetes patients, aims to integrate the mobile technology into LifeScan's glucose monitors.	MedDevice (2016)	Ι
J5	Nova Biomedical	The company announced today an exclusive agreement with Nova Biomedical to sell J & J's StatStrip, a blood glucose testing system, to hospitals in China for serving the fast-changing needs of emerging markets.	Prnewswire (2014)	Ш
J6	PayrollServe	PayrollServe was always knowledgeable and up-to-date about payroll matters, saving Johnson & Johnson Singapore time and energy.	Payroll-Serve (2016)	Ш

• Case of GE

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
G1	Arterys	GE Healthcare and Arterys have agreed to collaborate, incorporating Arterys' proprietary quantification and medical imaging technology with GE Healthcare's MR cardiac solutions. "Using the Arterys System together with cloud computation, the diagnoses can be made in a non-invasive MR imaging procedure that is quick, accessible and easy for both the patient and the radiologist."	Fierce-Biotech (2016)	Ι
G2	Getinge	General Electric (GE) Healthcare, in partnership with Swedish medical device firm Getinge Group, has launched a "highly flexible" angiography solution for hybrid operating room (OR) procedures. The solution combines GE's angiography system with Getinge Group Maquet brand's Magnus operating table system, which broadens the capabilities and services available in a hospital's Hybrid OR.	MedDevice (2016)	Ι
G3	Trice Imaging	A commercial partnership between GE Healthcare and Trice Imaging, Inc. will provide a new way for clinicians to connect with their patients. GE Healthcare's Ultrasound Women's Health product portfolio will be added by the new solution: cloud-based image sharing and diagnostic collaboration.	MedDevice (2016)	I
G4	Zenith Technologies	They will integrate process control systems and other supporting technologies with GE Healthcare's solutions for the biopharmaceutical manufacturing industry. "With Zenith Technologies' expertise in automation integration, we will be able to offer more choice to customers. This is ushers in a new era of opportunity for both companies," said Brendan O'Regan, founder at Zenith Technologies.	Zenith Technolo-gies (2016)	Ι

ID	Outsourcee	Summary of Events (Key Phrases from the Articles)	Source	Туре
G5	Capgemini	GE Healthcare today announced two systems integrators and seven leading-edge software vendors are planning to move their innovative solutions to the new GE Health Cloud. They will bring IT solutions for clinicians across a range of care areas.	MedDevice (2016)	Ι
G6	Barco Healthcare	"The GE Health Cloud will host powerful analytical tools and services that turn mountains of data into actionable insights," said CEO of GE Healthcare. "This is the future of healthcare: joining together to improve patient care and transform to value-based care."	MedDevice (2016)	Ι
G7	Tata Consultancy	"The industry needs disruptive technology as it navigates rising clinical, operational and financial pressure. It's a collaborative, healthcare-specific ecosystem that will solve healthcare's greatest challenges," said CEO of GE Healthcare IT.	MedDevice (2016)	Ι
G8	Mayo Clinic	"Research in this area aims to bring benefits of lower total costs, better image quality, and greater patient comfort. It's an attractive opportunity for collaboration," said Jim Davis, general manager of GE Healthcare's MRI business.	MedDevice (2016)	Ι
G9	CS Medical	"This agreement demonstrates both companies' continued commitment to minimizing Healthcare Acquired Infections. CS Medical provides healthcare professionals the solution to effectively reprocess the delicate ultrasound probe. GE's ongoing pledge to provide a variety of ultrasound disinfection solutions to help our customers improve patient outcomes."	CS Medical (2016)	п
G10	Masimo	"Masimo's suite of noninvasive and continuous measurements will further enhance the point-of-care clinical intelligence benefits already offered in GE Healthcare's patient monitors," said Matthias Weber, General Manager at GE Healthcare.	Masimo (2012)	Π
G11	SonoSim	"SonoSim's solution offers training in everything from the basics of scanning to hands-on simulations of real-world, pathologic cases from actual patients."	Business-Wire (2015)	П
G12	InTouch	Training doctors and nurses on all the machines General Electric manufactures is typically cumbersome and expensive, involving flying doctors to training sites or else sending experts out to hospitals. But a new partnership promises to bring remote education, training and technical support for a slew of GE imaging products.	Business Journal (2016)	Ш

References

- 1. Suh, C.; Lee, I. An empirical study on the manufacturing firm's strategic choice for sustainability in SMEs. *Sustainability* **2018**, *10*, 572.
- 2. Barney, J. Firm resources and sustained competitive advantage. J. Manag. 1991, 17, 99–120. [CrossRef]
- 3. Espino-Rodríguez, T.F.; Padrón-Robaina, V. A review of outsourcing from the resource-based view of the firm. *Int. J. Manag. Rev.* **2006**, *8*, 49–70. [CrossRef]
- 4. Hagedoorn, J. Inter-firm R&D partnerships: An overview of major trends and patterns since 1960. *Res. Policy* **2002**, *31*, 477–492.
- 5. Unal, B. The Role of Dynamic Capabilities in outsourcing sales and marketing functions: A resource-advantage perspective in the context of consumer packaged goods. Ph.D. Thesis, Georgia State University, Atlanga, GA, USA, 2011.
- 6. De Felice, F.; Petrillo, A.; Autorino, C. Development of a framework for sustainable outsourcing: Analytic Balanced Scorecard Method (A-BSC). *Sustainability* **2015**, *7*, 8399–8419. [CrossRef]
- Pellicelli, M. Gaining Flexibility and Innovation through Offshore Outsourcing. Sustainability 2018, 10, 1672. [CrossRef]

- 8. De Quinn, J.B. Outsourcing innovation: The new engine of growth. Sloan Manag. 2000, 41, 13.
- 9. Farughi, H.; Mostafayi, S. A hybrid approach based on ANP, ELECTRE and SIMANP metaheuristic method for outsourcing manufacturing procedures according to supply chain risks-Case study: A medical equipment manufacturer company in Iran. *Decis. Sci. Lett.* **2017**, *6*, 77–94. [CrossRef]
- 10. Merino, F. Offshoring, outsourcing and the economic geography of Europe. *Pap. Reg. Sci.* **2017**, *96*, 299–323. [CrossRef]
- 11. Dahlman, C.J. The problem of externality. J. Law Econ. 1979, 22, 141–162. [CrossRef]
- 12. Vivek, S.D.; Richey, R.G., Jr.; Dalela, V. A longitudinal examination of partnership governance in offshoring: A moving target. *J. World Bus.* **2009**, *44*, 16–30. [CrossRef]
- 13. Neves, L.W.D.A.; Hamacher, S.; Scavarda, L.F. Outsourcing from the perspectives of TCE and RBV: A multiple case study. *Production* **2014**, 24, 687–699. [CrossRef]
- 14. Van Hoek, R.I. The purchasing and control of supplementary third-party logistics services. *J. Supply Chain Manag.* **2000**, *36*, 14–26. [CrossRef]
- 15. Nooteboom, B. Governance and competence: How can they be combined? *Camb. J. Econ.* **2004**, *28*, 505–525. [CrossRef]
- 16. Ellram, L.M.; Tate, W.L.; Billington, C. Offshore outsourcing of professional services: A transaction cost economics perspective. *J. Oper. Manag.* **2008**, *26*, 148–163.
- 17. Verwaal, E.; Commandeur, H.; Verbeke, W. Value creation and value claiming in strategic outsourcing decisions: A resource contingency perspective. *J. Manag.* **2009**, *35*, 420–444. [CrossRef]
- 18. Holcomb, T.R.; Hitt, M.A. Toward a model of strategic outsourcing. J. Oper. Manag. 2007, 25, 464–481. [CrossRef]
- 19. Das, T.K.; Teng, B.-S. A resource-based theory of strategic alliances. J. Manag. 2000, 26, 31-61. [CrossRef]
- 20. Quinn, J.B.; Hilmer, F.G. Strategic outsourcing. MIT Sloan Manag. Rev. 1994, 35, 43.
- 21. Alegre, J.; Sengupta, K.; Lapiedra, R. Knowledge management and innovation performance in a high-tech SMEs industry. *Int. Small Bus. J.* **2013**, *31*, 454–470.
- 22. Chang, C.C. Exploring IT entrepreneurs' dynamic capabilities using Q-technique. *Ind. Manag. Data Syst.* **2012**, *112*, 1201–1216. [CrossRef]
- 23. Shang, S.S.; Lin, S.-F.; Wu, Y.-L. Service innovation through dynamic knowledge management. *Ind. Manag. Data Syst.* **2009**, *109*, 322–337. [CrossRef]
- 24. Troacă, V.-A.; Bodislav, D.-A. Outsourcing. The Concept. Theor. Appl. Econ. 2012, 6, 51.
- 25. McCarthy, I.; Anagnostou, A. The impact of outsourcing on the transaction costs and boundaries of manufacturing. *Int. J. Prod. Econ.* **2004**, *88*, 61–71. [CrossRef]
- 26. Lei, D.; Hitt, M.A. Strategic restructuring and outsourcing: The effect of mergers and acquisitions and LBOs on building firm skills and capabilities. *J. Manag.* **1995**, *21*, 835–859. [CrossRef]
- 27. Beaumont, N. Service level agreements: An essential aspect of outsourcing. *Serv. Ind. J.* **2006**, *26*, 381–395. [CrossRef]
- Dutta, A.; Roy, R. Offshore outsourcing: A dynamic causal model of counteracting forces. J. Manag. Inf. Syst. 2005, 22, 15–35. [CrossRef]
- 29. Feenstra, R.C.; Hanson, G.H. Ownership and control in outsourcing to China: Estimating the property-rights theory of the firm. *Q. J. Econ.* **2005**, *120*, 729–761.
- 30. Kennedy, J.F.; Holt, D.T.; Ward, M.A.; Rehg, M.T. The Influence of Outsourcing on Job Satisfaction and Turnover Intentions of Technical Managers. *Hum. Resour. Plan.* **2002**, *25*, 23–31.
- 31. Nam, K.; Rajagopalan, S.; Rao, H.R.; Chaudhury, A. A two-level investigation of information systems outsourcing. *Commun. ACM* **1996**, *39*, 36–45. [CrossRef]
- 32. Gilley, K.M.; Rasheed, A.A.; Al-Shammari, H. Research on Outsourcing: Theoretical Perspectives and Empirical Evidence. In *Global Outsourcing Strategies*; Routledge: Oxford, UK, 2017; pp. 41–56.
- 33. Bharadwaj, S.S.; Saxena, K.B.C.; Halemane, M.D. Building a successful relationship in business process outsourcing: An exploratory study. *Eur. J. Inf. Syst.* **2010**, *19*, 168–180. [CrossRef]
- 34. Williamson, O.E. The economic institutions of capitalism. Firms, markets, relational contracting. In *Das Summa Summarum des Management*; Springer: Berlin/Heidelberg, Germany, 2007; pp. 61–75.
- 35. Hashai, N. Focusing the high-technology firm: How outsourcing affects technological knowledge exploration. *J. Manag.* **2018**, *44*, 1736–1765. [CrossRef]
- 36. Williamson, O.E. Markets and hierarchies. New York 1973, 63, 316–325.

- 37. Coase, R.H. The nature of the firm. *Economica* 1937, 4, 386–405. [CrossRef]
- Rindfleisch, A.; Heide, J.B. Transaction cost analysis: Past, present, and future applications. J. Mark. 1997, 61, 30–54. [CrossRef]
- 39. Ray, G.; Barney, J.B.; Muhanna, W.A. Capabilities, business processes, and competitive advantage: Choosing the dependent variable in empirical tests of the resource-based view. *Strateg. Manag. J.* **2004**, *25*, 23–37. [CrossRef]
- 40. Stalk, G.; Evans, P.; Shulman, L.E. Competing on capabilities: The new rules of corporate strategy. *Harv. Bus. Rev.* **1992**, *70*, 57–69. [PubMed]
- 41. Silverman, B.S. Technological resources and the direction of corporate diversification: Toward an integration of the resource-based view and transaction cost economics. *Manag. Sci.* **1999**, *45*, 1109–1124. [CrossRef]
- 42. Foss, N.J. The resource-based perspective: An assessment and diagnosis of problems. *Scand. J. Manag.* **1998**, *14*, 133–149. [CrossRef]
- 43. Penrose, E.; Penrose, E.T. The Theory of the Growth of the Firm; Oxford University Press: Oxford, UK, 2009.
- 44. Hunt, S.D.; Lambe, C.J.; Wittmann, C.M. A theory and model of business alliance success. *J. Relatsh. Mark.* **2002**, *1*, 17–35. [CrossRef]
- 45. Madhok, A. Cost, value and foreign market entry mode: The transaction and the firm. *Strateg. Manag. J.* **1997**, *18*, 39–61. [CrossRef]
- 46. Ireland, R.D.; Covin, J.G.; Kuratko, D.F. Conceptualizing corporate entrepreneurship strategy. *Entrep. Theory Pract.* **2009**, *33*, 19–46. [CrossRef]
- 47. Espino-Rodríguez, T.F.; Chun-Lai, P.; Gil-Padilla, A.M. Does outsourcing moderate the effects of asset specificity on performance? An application in Taiwanese hotels. *J. Hosp. Tour. Manag.* **2017**, *31*, 13–27. [CrossRef]
- 48. Peteraf, M.A. Commentary: The two schools of thought in resource-based theory: Definitions and implications for research. *Adv. Strateg. Manag.* **1994**, *10*, 153–158.
- 49. Fahy, J.; Smithee, A. Strategic marketing and the resource based view of the firm. *Acad. Mark. Sci. Rev.* **1999**, *10*, 1–21.
- 50. Teece, D.J.; Pisano, G.; Shuen, A. Dynamic capabilities and strategic management. *Strateg. Manag. J.* **1997**, *18*, 509–533. [CrossRef]
- 51. Eisenhardt, K.M.; Martin, J.A. Dynamic capabilities: What are they? *Strateg. Manag. J.* **2000**, *21*, 1105–1121. [CrossRef]
- 52. Paek, B.; Lee, H. Strategic entrepreneurship and competitive advantage of established firms: Evidence from the digital TV industry. *Int. Entrep. Manag. J.* **2018**, *14*, 883–925. [CrossRef]
- 53. Zollo, M.; Winter, S.G. Deliberate learning and the evolution of dynamic capabilities. *Organ. Sci.* **2002**, *13*, 339–351. [CrossRef]
- 54. Song, L.; Jing, L. Strategic orientation and performance of new ventures: Empirical studies based on entrepreneurial activities in China. *Int. Entrep. Manag. J.* **2017**, *13*, 989–1012. [CrossRef]
- 55. Griffith, D.A.; Harvey, M.G. A resource perspective of global dynamic capabilities. *J. Int. Bus. Stud.* **2001**, *32*, 597–606. [CrossRef]
- 56. Butler, B.; Soontiens, W. Offshoring of higher education services in strategic nets: A dynamic capabilities perspective. *J. World Bus.* **2015**, *50*, 477–490. [CrossRef]
- 57. Mahroum, S.; Al-Saleh, Y. Towards a functional framework for measuring national innovation efficacy. *Technovation* **2013**, *33*, 320–332. [CrossRef]
- 58. Dyer, J.H.; Nobeoka, K. Creating and managing a high-performance knowledge-sharing network: The Toyota case. *Strateg. Manag. J.* **2000**, *21*, 345–367. [CrossRef]
- 59. Unal, B.; Donthu, N. Role of absorptive capabilities in outsourcing the headquarters selling task in the United States. *Ind. Mark. Manag.* **2014**, *43*, 1079–1084. [CrossRef]
- 60. Yin, R.K. *Case Study Research and Applications: Design and Methods*; Sage Publications: Thousand Oaks, CA, USA, 2017.
- 61. Dodgson, M. Organizational learning: A review of some literatures. *Organ. Stud.* **1993**, *14*, 375–394. [CrossRef]
- 62. Bos, W.; Tarnai, C. Content analysis in empirical social research. *Int. J. Educ. Res.* **1999**, *31*, 659–671. [CrossRef]

- 63. Krippendorff, K. *Content Analysis: An Introduction to its Methodology*; Sage Publications: Thousand Oaks, CA, USA, 2018.
- 64. Wernerfelt, B. A resource-based view of the firm. Strateg. Manag. J. 1984, 5, 171–180. [CrossRef]
- 65. Wu, L.-Y. Applicability of the resource-based and dynamic-capability views under environmental volatility. *J. Bus. Res.* **2010**, *63*, 27–31. [CrossRef]
- 66. Zott, C. Dynamic capabilities and the emergence of intraindustry differential firm performance: Insights from a simulation study. *Strateg. Manag. J.* **2003**, *24*, 97–125. [CrossRef]
- 67. Belso-Martínez, J.A.; Xavier Molina-Morales, F.; Mas-Verdu, F. Clustering and internal resources: Moderation and mediation effects. *J. Knowl. Manag.* **2011**, *15*, 738–758. [CrossRef]
- 68. Fernández-Mesa, A.; Alegre-Vidal, J.; Chiva-Gómez, R.; Gutiérrez-Gracia, A. Design management capability and product innovation in SMEs. *Manag. Decis.* **2013**, *51*, 547–565. [CrossRef]
- 69. Lavie, D. The competitive advantage of interconnected firms: An extension of the resource-based view. *Acad. Manag. Rev.* **2006**, *31*, 638–658. [CrossRef]
- 70. Quélin, B. Core competencies, R&D management and partnerships. Eur. Manag. J. 2000, 18, 476–487.
- Fine, C.H. Clockspeed: Winning Industry Control in the Age of Temporary Advantage; ReadHowYouWant.com, 2010; ISBN 978-0738201535. Available online: https://books.google.com.hk/books/about/Clockspeed.html? id=zG5AExER6mwC&redir_esc=y (accessed on 21 August 2019).
- 72. Prahlad, C.; Hamel, G. The core competence of the corporation. Harv. Bus. Rev. 1990, 68, 79–91.
- 73. Chang, Y.-C.; Chang, H.-T.; Chi, H.-R.; Chen, M.-H.; Deng, L.-L. How do established firms improve radical innovation performance? The organizational capabilities view. *Technovation* **2012**, *32*, 441–451. [CrossRef]
- 74. Li, D.-Y.; Liu, J. Dynamic capabilities, environmental dynamism, and competitive advantage: Evidence from China. *J. Bus. Res.* **2014**, *67*, 2793–2799. [CrossRef]
- 75. Nicholls-Nixon, C.L.; Woo, C.Y. Technology sourcing and output of established firms in a regime of encompassing technological change. *Strateg. Manag. J.* **2003**, 24, 651–666. [CrossRef]
- 76. Roy, K.; Khokhle, P.W. Integrating resource-based and rational contingency views: Understanding the design of dynamic capabilities of organizations. *Vikalpa* **2011**, *36*, 67–76. [CrossRef]
- 77. Teece, D.J. Explicating dynamic capabilities: The nature and microfoundations of (sustainable) enterprise performance. *Strateg. Manag. J.* **2007**, *28*, 1319–1350. [CrossRef]
- 78. Sanchez, R.; Heene, A. Competence-Based Strategic Management: Concepts and Issues for Theory, Research, and *Practice*; Wiley: Hoboken, NJ, USA, 1997.
- 79. Mahnke, V. The process of vertical dis-integration: An evolutionary perspective on outsourcing. *J. Manag. Gov.* **2001**, *5*, 353–379. [CrossRef]
- 80. Zhan, W.; Chen, R.R. Dynamic capability and IJV performance: The effect of exploitation and exploration capabilities. *Asia Pac. J. Manag.* **2013**, *30*, 601–632. [CrossRef]
- 81. Hobday, M.; Davies, A.; Prencipe, A. Systems integration: A core capability of the modern corporation. *Ind. Corp. Chang.* **2005**, *14*, 1109–1143. [CrossRef]
- 82. Jiang, W.; Mavondo, F.T.; Matanda, M.J. Integrative capability for successful partnering: A critical dynamic capability. *Manag. Decis.* 2015, 53, 1184–1202. [CrossRef]
- 83. Lema, R.; Quadros, R.; Schmitz, H. Reorganising global value chains and building innovation capabilities in Brazil and India. *Res. Policy* **2015**, *44*, 1376–1386. [CrossRef]
- 84. Zhang, M.; Guo, H.; Huo, B.; Zhao, X.; Huang, J. Linking supply chain quality integration with mass customization and product modularity. *Int. J. Prod. Econ.* **2019**, 207, 227–235. [CrossRef]
- 85. Di Gregorio, D.; Musteen, M.; Thomas, D.E. Offshore outsourcing as a source of international competitiveness for SMEs. *J. Int. Bus. Stud.* **2009**, *40*, 969–988. [CrossRef]
- 86. Holzweber, M.; Mattsson, J.; Chadee, D.; Raman, R. How dynamic capabilities drive performance in the Indian IT industry: The role of information and co-ordination. *Serv. Ind. J.* **2012**, *32*, 531–550. [CrossRef]
- 87. Kazozcu, S.B. Role of strategic flexibility in the choice of turnaround strategies: A resource based approach. *Proc. Soc. Behav. Sci.* **2011**, 24, 444–459. [CrossRef]
- 88. Lin, Y.; Wu, L.-Y. Exploring the role of dynamic capabilities in firm performance under the resource-based view framework. *J. Bus. Res.* **2014**, *67*, 407–413. [CrossRef]
- 89. Enyinda, C.; Briggs, C.; Bachkar, K. Managing risk in pharmaceutical global supply chain outsourcing: Applying analytic hierarchy process model. In Proceedings of the ASBBS Annual Conference, Las Vegas, NV, USA, 19–22 February 2009.

- Persson, G.; Virum, H. Growth strategies for logistics service providers: A case study. *Int. J. Logist. Manag.* 2001, 12, 53–64. [CrossRef]
- 91. Dixon, S.; Meyer, K.; Day, M. Building dynamic capabilities of adaptation and innovation: A study of micro-foundations in a transition economy. *Long Range Plan.* **2014**, *47*, 186–205. [CrossRef]
- 92. Cui, Y.; Jiao, H. Dynamic capabilities, strategic stakeholder alliances and sustainable competitive advantage: Evidence from China. *Corp. Gov. Int. J. Bus. Soc.* **2011**, *11*, 386–398. [CrossRef]
- 93. Swafford, P.M.; Ghosh, S.; Murthy, N. The antecedents of supply chain agility of a firm: Scale development and model testing. *J. Oper. Manag.* 2006, *24*, 170–188. [CrossRef]
- 94. Stalk, G., Jr.; Hout, T.M. Competing against time. Res. Technol. Manag. 1990, 33, 19–24. [CrossRef]
- 95. Yung, I.-S.; Lai, M.-H. Dynamic capabilities in new product development: The case of Asus in motherboard production. *Total Qual. Manag. Bus. Excell.* **2012**, *23*, 1125–1134. [CrossRef]
- 96. George, G.; McGahan, A.M.; Prabhu, J. Innovation for inclusive growth: Towards a theoretical framework and a research agenda. *J. Manag. Stud.* **2012**, *49*, 661–683. [CrossRef]
- 97. Schneckenberg, D.; Truong, Y.; Mazloomi, H. Microfoundations of innovative capabilities: The leverage of collaborative technologies on organizational learning and knowledge management in a multinational corporation. *Technol. Forecast. Soc. Chang.* **2015**, *100*, 356–368. [CrossRef]
- World Health Organization. Systematic Review of Needs for Medical Devices for Ageing Populations. Available online: https://www.who.int/medical_devices/systematic_Review_needs.pdf (accessed on 2 August 2019).
- 99. Maresova, P.; Penhaker, M.; Selamat, A.; Kuca, K. The potential of medical device industry in technological and economical context. *Ther. Clin. Risk Manag.* **2015**, *11*, 1505. [PubMed]
- Research, G.V. Medical Device Outsourcing Market Size Worth \$210.5 Billion By 2026. Available online: https://www.grandviewresearch.com/press-release/global-medical-device-outsourcing-market (accessed on 5 April 2019).
- 101. Loecher, U. Small and medium-sized enterprises–delimitation and the European definition in the area of industrial business. *Eur. Bus. Rev.* 2000, *12*, 261–264. [CrossRef]
- OECD. Small and Medium-Sized Enterprises (SMES). Available online: https://stats.oecd.org/glossary/detail. asp?ID=3123 (accessed on 5 January 2019).
- 103. World Trade Organization. Micro, Small and Medium-Sized Enterprises. Available online: https://www.wto. org/english/thewto_e/minist_e/mc11_e/briefing_notes_e/bfmsmes_e.htm (accessed on 5 May 2019).



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).